Author’s response to reviews

Title: The mutation spectrum in familial versus sporadic congenital cataract based on next-generation sequencing

Authors:

Fan Fan (fanfan432121@hotmail.com)
Yi Luo (yeeluo1106@163.com)
Jihong Wu (jihongwu@fudan.edu.cn)
Chao Gao (gaochao_1990@126.com)
Xin Liu (liuxin0545032@sina.com)
Hengjun Mei (13301050285@fudan.edu.cn)
Xiyue Zhou (13301050257@fudan.edu.cn)

Version: 5 Date: 20 Jun 2020

Author’s response to reviews:

BMC Ophthalmology

Dear Editor Jashan P Abraham,

Thank you very much for your consideration of our manuscript, “The mutation spectrum in familial versus sporadic congenital cataract based on next-generation sequencing” (BOPH-D-20-00148R4).

We have provided point-by-point responses to the comments below.

We look forward to receiving your final decision.

Yours sincerely,

Yi Luo
Editor Comments:

1. Table
Please ensure Table S1 and Table 4 are cited in the body of text in the manuscript
Reply: Table 4 were referred in Discussion section (Line 161, Page4). We apologize for missing Table S1 citation in text. It has been cited in Method section (Line 97, Page 3).

2. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Reply: We ensured that all the upload files were clean version and they were separated as manuscript, tables, figure and additional file.